	Date
	Bench
	Problem
	Type
	Application/
soft
	Action/advise

	Tuesday _23_N
	TBB2

	MB3374: trigger during SPA at 1 kA. 
	MTM
	
	Voltage level of the D1-D2 card still for SSS (100 mV).
Voltage levels have to be better adjusted.

	Wednesday_24_A
	TBC2

	Insulation fault on cold initial before test connected to CFB temperature vs ground
	MTM
	
	Richard found the origin :  humidity in the CFB current leads.

	Wednesday_24_A
	TBA2

	MB1084: CDAP failed at dipole -quench heaters Vs ground.
	MTM
	
	Richard found that the new installation for AC cables placed near the support of the main cables for the current leads are inducing a contact.
Wires from the current leads to new connector were disconnected and test launched again

	Wednesday _24_A
	TBC2

	For MB3384 the manual MIITs for 1.5 kA quench heater discharges could not be registered.
	AB/CO
	AQA
	

	Wednesday _24_N
	TBA2

	MB1084: PTC10 - AQA alarm - DirVT voltage peak exceeded 200V.
	MTM
	
	Giorgio consulted. Check the curves and give the OK to continue

	Thursday_25_N
	TBE2


	No trigger for hf DAQ.
	AB/CO
	hf
	Giorgio changed the CPU and the last DAQ cards.

	Thursday_25_N
	TBE2

	HWREC is not working.
	?
	HWREC
	

	Friday_26_A
	TBD1


	SSS180 cooled down start. Anticryostat temperature went down.
	MTM
	
	P.Legrand checked and the resistances are working. Cold point coming from the magnet. Warm up the magnet. Disconnected and replaced.

	Friday_26_A
	TBF2


	trigger on potaim cards Vsum & Vtotal
	MTM
	
	Giorgio changed the potaim-card seven.

	Saturday_27_M
	TBC2


	ICS carried out WP04 instead of WP07. The original WP04 was overwritten by new file..
	ICS
	
	The test name was changed in htm file and the file name was corrected in HV MTF Report folder.

	Saturday_27_M
	TBF2


	Power Supply for minimum energy (PT 10) is not working.
No voltage reading.
	MTM
	
	Xavier forgot to restore the voltage readout cable in proper position. Giorgio fixed the problem.

	Saturday_27_M
	TBE1


	Leak suspected on MB3363.
	
	
	Cryo team informed that quench can be performed only after their inspection/checking.

	Sunday_28_M
	TBA1


	MB2067: MinNRJ to warm up. Alarm Iearth>10 mA (12 mA).
	MTM
	AQA
	Giorgio contacted. 

	Sunday_28_M
	TBC2


	MB3388: HV test at warm: Dipole vs ground failed.
	ICS
	
	ICS will open the manchette and check the connections Bad connection found.. 

	Sunday_28_A
	TBB2


	.ub file generated and SPA, no trigger recieved, Cryo OK lost 
	MTM/ACR?
	
	ACR team said that a quench signal was received..

	Monday_29_M
	All


	MTF was not working. No connection to DB.
	TS/CE
	
	


